Author’s response to reviews

Title: Iatrogenic left anterior descending artery stenosis early after aortic valve replacement presenting with T-wave-pseudonormalization - a case report

Authors:

Stefan Schwarz (stef.schwarz@yahoo.de)
Hermann Blessberger (hermann.blessberger@akh.linz.at)
Jürgen Kammler (juergen.kammler@akh.linz.at)
Christoph Gross (christoph.gross@akh.linz.at)
Clemens Steinwender (clemens.steinwender@akh.linz.at)

Version: 3 Date: 12 February 2015

Author’s response to reviews: see over
IATROGENIC LEFT ANTERIOR DESCENDING ARTERY STENOSIS EARLY AFTER AORTIC VALVE REPLACEMENT PRESENTING WITH T-WAVE-PSEUDONORMALIZATION - A CASE REPORT

Stefan Schwarz¹, Hermann Blessberger¹, Jürgen Kammler¹, Christoph Gross², Clemens Steinwender¹

¹1ST MEDICAL DEPARTMENT - CARDIOLOGY, JOHANNES KEPLER UNIVERSITY SCHOOL OF MEDICINE, KRANKENHAUSSTRASSE 9, 4020 LINZ, AUSTRIA
²DEPARTMENT OF CARDIOTHORACIC SURGERY, JOHANNES KEPLER UNIVERSITY SCHOOL OF MEDICINE, KRANKENHAUSSTRASSE 9, 4020 LINZ, AUSTRIA

Educational impact:
Both iatrogenic coronary artery stenosis early after aortic valve replacement and pseudonormalization of inverted T-waves in acute ischemia are rare but well-recognized findings. We performed a PubMed-related search using the keywords mentioned on page 2, and to our knowledge, no case report covering the coincidence of both phenomena has been reported so far. In the reported case, the history of normal preoperative coronary angiography and ECG recording during acute chest pain without pathological findings provide possible pitfalls to overlook an unstable, potentially life-threatening condition. So we think that this case report may be helpful in detecting a small number of patients at high risk that should lead to admission to the ICU followed by early invasive management.

Statement:
The manuscript is original, and any part of it has not been and will not be submitted elsewhere for publication. Each author has contributed to, read and approved the final version of the manuscript. The authors declare that there is no known conflict of interest.